\documentstyle[12pt]{article}
\pagestyle{empty} % put % in col 1 if you want page numbers
\begin{document}

\noindent
Math 325

\noindent
Spring, 2001

\bigskip

\begin{center}
{\bf ASSIGNMENT 7, due Friday, March  30}
\end{center}

\bigskip
\textbf{Reminder:} Project 1 is due Wednesday, April 4.

\bigskip
Read Boyce and DiPrima, Sections 10.1-10.2 and {\it
  Differential Equations with Maple}, Chapter 13 (and reread Chapter
12 if necessary).

\medskip

Do Problem Set F \#5,13 in {\it Ordinary
Differential Equations with Maple}.  
You must use Maple for these problems.
For this part of the assignment, you may work in groups, following the same
rules as for Assignment 2.
If you do work in a group, turn these problems in separately from the
rest of the assignment.

\medskip

Do  p. 547 \#1,3,4,11,13,14,16
 
\end{document}

